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SHOW REPORT – ‘ONE DAY, MAYBE’

PERFORMANCE : 4							DATE : 03.09.17

SHOW 1 : 13:16		ATTN. 39
SHOW 2 : 13:48		ATTN. 38
SHOW 3 : 14:23		ATTN. 39
SHOW 4 : 15:15		ATTN. 38
SHOW 5 : 15:48		ATTN. 34
SHOW 6 : 16:28		ATTN. 39

NOTES

ONE :
1. Delayed start due to LX issue on the High Street
2. Virtual Reality 1 unit had a screen glitch

TWO :
1. One claustrophobic patron was escorted past the Maze.

THREE :
1. One patron on crutches was assisted through the Maze. Mx Hirst checked she was able to manage stairs before she was taken through the car park.

FOUR:
1. A magnet in the mini kabuki failed. Kabuki drop still worked. Magnet to be replaced.
2. The projector in the main installation was flashing – issue resolved by Mr Foley.

FIVE:
1. Same as Four 

SIX:
1. The lift went out of service. An engineer was called but the problem was fixed by Mr Franklin at 16:48. Mx Hirst & Foley now have instructions on how to fix it which will allow for swifter response in future.
2. Due to lift issue, a patron in a mobility scooter was unable to join the group for the final image due to the lift being out of service. Another patron from the previous group had to leave as an injured knee prevented her from using the stairs. Lift maintenance check will be out in place during the busier weekend periods.
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SHOW REPORT – ‘ONE DAY, MAYBE’

PERFORMANCE : 5							DATE : 03.09.17

SHOW 1 : N/A		ATTN. N/A
SHOW 2 : 18:30		ATTN. 19
SHOW 3 : 19:01		ATTN. 17
SHOW 4 : 20:15		ATTN. 25 
SHOW 5 : N/A		ATTN. N/A
SHOW 6 : N/A		ATTN. N/A

NOTES

TWO :
1. Initially there was no VR working but it was resolved. The VR assistants from the ground floor will be issued with further instructions on Tuesday.
FOUR:
1. The Bambi app is going to be put on to more iPads in case the ones in the unit fail.

iPads

The charging issue has been resolved, allowing for full compliment of i-pads if several shows back to back are sold out.

3rd Floor Reset

On average the reset took between eight and 10 minutes. The assistance from the performers made a big difference and was appreciated.

Costume Reset

Ms Ko was able to leave the show around 4pm as the SM team were coping well. 

Management of Jesa & Cells

To be considered on Tuesday. Staffing needs to be redeployed to make these spaces easier to run.

Other

1. The main projector needed to be reconfigured after the HDMI cable was swapped to solve the flashing problem. 
2. After the audience file into the High Street and the door is locked Mr Lavelle only just has enough time to reach the cue position for Q3. Please can the cast be made aware of this so they do not speed up in that moment before the MD’s speech.
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